Author's response to reviews

Title: Phenethyl isothiocyanate upregulates death receptors 4 and 5, sensitizes TRAIL-induced apoptosis, and inhibits proliferation in human cancer stem-like cells

Authors:

DAN WANG (DanWang@sdsstate.edu)
BIJAYA UPADHYAYA (bijaya.upadhyaya@sdsstate.edu)
YI LIU (yi.liu9845@sdsstate.edu)
DAVID KNUDSEN (David.Knudsen@sdsstate.edu)
MOUL DEY (MOUL.DEY@SDSTATE.EDU)

Version: 4 Date: 2 May 2014

Author's response to reviews:

Dear Editor,

We are submitting a revised version of the manuscript reflecting following changes:

1. Line numbers are now added in the main text.

Thank you
Sincerely
Moul Dey